Author's response to reviews

Title: Deregulation of microRNAs Let-7a and miR-21 mediate aberrant STAT3 signaling during Human Papillomavirus-induced Cervical Carcinogenesis: Role of E6 Oncoprotein

Authors:

Gauri Shishodia (gaurishishodia@gmail.com)
Gaurav Verma (gverma.gaurav@gmail.com)
Yogesh Srivastava (yansh111@gmail.com)
Ravi Mehrotra (rm8509@gmail.com)
Bhudev C. Das (bcdas48@hotmail.com)
Alok C. Bharti (bhartiac@icmr.org.in)

Version: 5 Date: 13 November 2014

Author's response to reviews: see over
Dear Dr. Polterauer,

Thank you very much for accepting the manuscript. Please find enclosed the revised manuscript as per your request.

Best wishes

AC Bharti

Editor's Request:

1. Please format your abstract according to our guidelines
   Response: Included in the revised M/s.

2. Please include a conclusion section according to our guidelines:
   Response: Included in the revised M/s.

3. Competing interests: Please include a 'Competing interests' section between the Conclusions and Authors'
   Response: Included in the revised M/s.

4. Please can you minimize using abbreviations in your acknowledgements section.
   Response: Abbreviations have been removed as suggested.